Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/rser





CrossMark

Andreas Uihlein*, Davide Magagna

European Commission, Joint Research Centre (JRC), Institute for Energy and Transport (IET), P.O. Box 2, 1755 ZG Petten, The Netherlands

A R T I C L E I N F O

ABSTRACT

Article history: Received 5 January 2015 Received in revised form 23 November 2015 Accepted 27 December 2015 Available online 15 January 2016

Keywords: Ocean energy Ocean wave Tidal current Research activities R&D The oceans of the earth offer vast amounts of renewable energy. Technologies to harness the power of the seas are at an early stage of development. Even the most advances technologies, namely tidal current and ocean wave still face considerable barriers and many obstacles remain. Research, development and innovation can help overcome those barriers. This review provides an overview over the current state of research in the field of ocean energy. In particular, the authors focus on research beyond technology or technological improvements. This article also highlights areas where research gaps exists and where future research efforts should be directed to.

© 2016 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

Contents

1.	Intro	luction
2.	Resou	Irce assessment and forecasting
	2.1.	Ocean wave
		2.1.1. Resource assessment
		2.1.2. Forecasting
	2.2.	Tidal current
		2.2.1. Resource assessment
		2.2.2. Forecast
	2.3.	Future research
3.	Envir	onmental impacts
	3.1.	Life Cycle Assessment
	3.2.	Environmental impact assessment and strategic environmental assessment
		3.2.1. Tidal current
		3.2.2. Ocean wave
	3.2.	Future research
4.	Socio	-economic impacts
	4.1.	Costs
	4.2.	Social impacts
	4.3.	Cost-benefit analysis
	4.4.	Future research
5.	Grid i	integration
	5.1.	Variability of resources
	5.2.	Grid connection and grid codes

* Corresponding author. Tel.: +31 224 56 5123.

E-mail address: andreas.uihlein@ec.europa.eu (A. Uihlein).

http://dx.doi.org/10.1016/j.rser.2015.12.284

1364-0321/© 2016 The Authors. Published by Elsevier Ltd. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/by-nc-nd/4.0/).

	5.3.	Power quality and control	1076
	5.4.	Future research	1077
6.	Installa	ation, operation and maintenance	1077
	6.1.	Array configuration	1077
	6.2.	Installation	1077
	6.3.	Operation and maintenance	1077
	6.4.	Future research	1078
7.	Regula	ntory and legal affairs	1078
	7.1.	Legal framework	1078
	7.2.	Maritime and spatial planning	1078
	7.3.	Future research	1078
8.	Conclu	isions	1078
9.	Disclai	iner	1079
Refe	erences		1079

1. Introduction

The oceans of the earth represent a vast source of renewable energy. In general, ocean energy can be divided into six types of different origin and characteristics: ocean wave, tidal range, tidal current, ocean current, ocean thermal energy, and salinity gradient [1-3].

Currently, all ocean energy technologies except tidal range can be considered at an early stage of development from conceptual up to demonstration stage [2]. Ocean wave and tidal current energy are the two types of ocean energy which are most advanced and are expected to contribute significantly to the supply of energy in future [2]. Thus, the authors will focus on those two types ocean energy in this paper.

The ocean energy industry has made significant progress in recent years but is still at very early stage with some advanced prototypes that are currently being tested [4]. Existing challenges include further development of the technology to prove reliability and robustness and to reduce costs but also deployment and risk reduction. This is reflected in the current research themes funded e.g. by the EU with 68% of the funds being directed to technology development (Fig. 1). However, other not technology-related knowledge gaps and barriers exist [4].

The aim of this review is to provide an overview over the current state of research in the field of wave and tidal current energy. Further research and innovation in the area of technology is the prerequisite to tap the full potential of ocean energy. According to [5], technological barriers represent the most important issue that the ocean energy sector needs to address in the short-medium term. Priority topics include e.g. technology advancement, reliability demonstration,





sub-system development and optimisation, pre-commercial array sea trial and demonstration, predictive maintenance systems, and array electrical systems [4].

However, also other areas require attention but they are tackled to a minor extend partly. This article will focus on research beyond technology or technological improvements (for those, the authors would like to refer to technology-oriented studies and reports such as [4,6–8]) and identify areas where research gaps exists and where future research efforts should be directed to. The review is based on a literature review and desk-based research. Areas that will be covered in greater detail have been identified according to [2,7,9].

The article is structured as follows: Section 2 presents the state-of-research of resource assessment and forecasting; Section 3 will present research on environmental impacts. Section 4 covers socio-economic impacts and Section 5 grid integration. Section 6 discusses the current literature on array configuration, installation, operation and maintenance while in Section 7; some relevant works on regulatory and legal affairs will be presented. Section 8 contains some conclusions.

2. Resource assessment and forecasting

An important initial step towards market deployment of ocean energy is the characterisation and mapping of ocean energy resources. The assessment of wave energy resources includes the identification of areas with high wave energy, the quantification of average energy resources (e.g. total annual wave energy) and the description of the resource by using parameters such as significant wave height, wave energy period and mean wave direction [10]. Precise estimates and description of available wave energy resources at high spatial and temporal resolution are needed for proper planning and the optimisation of the design of ocean energy converters [11,12]. This will help to optimise device performance in terms of power produced. For example, the power output of a Oscillating Water Column device at a certain location has been studied [13]. The current state of technology development will determine how much of the resource can be exploited with the main technical parameters to be improved being device efficiency and capacity factor [10,14] . Reducing uncertainties concerning the available resources will also increase the confidence of investors as it allows a better determination of the value of investments and minimising risks [15,16]. In the following sections, an overview of current state of wave and tidal current energy resource assessment and forecasting is presented.

2.1. Ocean wave

2.1.1. Resource assessment

During the last years, ocean wave energy resources have been assessed for various regions in the world. The first wave energy resource assessments have been made using buoy data limited to local conditions [12]. The second generation of assessments included buoy data in combination with deep water numerical models which can assess offshore wave resources which helped overcoming the limitations of first generation assessments, namely the limited time period of the buoy measurements and the uncertainties of extrapolating local data to other locations [12]. Recent tools incorporate radar measurements and allow modelling wave generation and propagation also in coastal regions [11,17]. Usually, wind and bathymetry data are used as an input for such models. Typical output parameters are: significant wave height, mean wave period, peak wave period, and mean wave direction.

Ocean wave energy resource assessments have been performed on global, regional and local level. In Table 1, a non-exhaustive list of wave energy resource assessments is presented. Other coasts and areas that have been studied include California, Argentina, Australia, Portugal, Sweden, Korea, Spain, Iran, and the Atlantic coast of the United States [18].

2.1.2. Forecasting

Wave forecasting is performed by statistical techniques or physics-based models [19,20]. An example for a wave forecast system based on physical models is the wave model WAM that is incorporated into the Integrated Forecast System (IFS) of the European Centre for Medium-Range Weather Forecasts (ECMWF) [19]. The ECMWF model forecast includes up to a maximum of 48 h ahead in 3 h steps. Similarly, the National Oceanic and Atmospheric Administration (NOAA) and the National Weather Service (NWS) use WAVEWATCH-III based on wind information from the Global Data Assimilation Scheme (GDAS). WAVEWATCH-III produces forecasts of up to 180 h in 3 h steps. Other wave forecast include regional forecasts for the Gulf of Mexico (using SWAN and wind forecasts from NOAA) and forecast from private companies such as Oceanweather which uses a privately developed spectral wave model. Statistical approaches include neural networks, regressionbased techniques and genetic programming [20–25]. Some authors have tried to combine both methods [19]. It seems that statistical models are more accurate for short time horizons (up to 6 h) while physics-based model perform better for longer time horizons and a combination of both methods leads to more accurate results [19]. Statistical models might be sufficient for to be used for electricity utilities for generating and trading.

2.2. Tidal current

2.2.1. Resource assessment

Similar to wave energy, tidal current energy resources have been assessed since a number of years. Often, direct measurements have been performed on-site. Since some years, 2D and 3D modelling techniques have been applied to assess tidal current energy resources by modelling current velocities. More recent publications assess also the hydrodynamic effects of power extraction and consider for example change to the flow field, change in water surface elevation, or disturbances in tidal dynamics [26,27].

According to [14], tidal current energy is calculated as function of seawater density velocity, velocity availability factor, neap/spring factor, and peak spring-tide velocity. However, it is not possible to convert all tidal current energy power due to Betz' law and mechanical losses in the turbines. These limitations are accounted via the power coefficient.

Tidal energy resource assessments have been performed for many regions and coastal areas of the world. Table 2 shows examples for studies in the field.

2.2.2. Forecast

Tidal current forecasts are usually readily available [28]. For example, NOAA Current Predictions allows forecasts up to 48 hours, one week and annual predictions, which are available online.

The German Federal Maritime and Hydrographic Agency provides current predictions of up to 3 days [29]. The model used is a 3D model that takes into account meteorological forecasts for the North Sea and Baltic Sea provided by the German Weather Service (DWD) [30], tides and external surges entering the North Sea from the Atlantic, as well as river runoff from the major rivers. The

Table 1

Examples of ocean wave energy resource assessments.

Region	Parameter- s ^a	Modelling approach and tools	Refer- ence
Global	H _S , T _m , T _p , P	WAVEWATCH-III using FNL wind data from the Global Forecast System (GFS). 19 buoys and satellite altimetry data used for model validation	[11]
Global	Н _S , Т _р , Р	WAVEWATCH-III based on wind fields input from NCEP. Spatial resolution of $1.25^{\circ} \times 1.0^{\circ}$, 3 h intervals	[17]
Global	$H_{\rm S}, T_{\rm p}, \theta_{\rm m}, P$	WAVEWATCH-III. Spatial resolution 30', temporal resolution 3 h	[15]
UK (Cornwall)	$H_{\rm S}, T_{\rm e}, \theta_{\rm m}, P$	Wave model SWAN, wave and wind input from ECMWF. Grid resolution $1.5^{\circ} \times 1.5^{\circ}$, 3 h temporal resolution. Validation with buoys measurements	[133]
Eastern Mediterranean and Aegean Seas	H _s , T _m , P	Spectral wave model (hourly) using wind data from ECMWF. Model calibrated using the wave mea- surements conducted at three different stations	[134]
Western French coast	H _s , T _p , T _m , P	WAVEWATCH III providing boundary conditions for SWAN. Wind input from ECMWF, compared and validated with measurements from three buoys	[12]
Azores islands	$H_{\rm S}$, $T_{\rm e}$, $\theta_{\rm m}$, P	Analysis of wave climate using remotely sensed and historical data. Model based on WAM for wave generation to provide boundary conditions for SWAN	[135]
Hawaii	H _S , T _e , P	WAVEWATCH-III model based on FNL wind data for far field, Hawaii WAVEWATCH-III model with high- resolution winds from WRF for local wave processes. SWAN model for coastal waters. Validation with measurements from satellites and buoys	[136]
Galicia	$H_{\rm S}, T_{\rm e}, \theta_{\rm m}, P$	Offshore buoy used as boundary condition. Wave propagation to coastline modelled with SWAN. Validation with offshore and coastal wave buoys	[10]
Black Sea	H _S , T _e , P	6-h wind fields from ECMWF as input for SWAN (1.3 \times 1.83 km). Measurements from a wave buoy for verification	[18]
India	H _S , T _p	WAVEWATCH-III model combined with SWAN. IFREMER/CERSAT blended winds used as input. 1.133° \times 1.133° spatial grid with 0.5 h temporal resolution. Validation with buoy measurements.	[137]

^a $H_{\rm S}$: significant wave height; $T_{\rm m}$: mean wave period; $T_{\rm p}$: peak wave period; $T_{\rm e}$: wave energy period; $\theta_{\rm m}$: mean wave direction; P: wave power.

Table 2

Examples of tidal current energy resource assessments.

Region	Parameters	Modelling approach and tools	Refer- ence
Ireland	Peak spring-tide velocity	2D depth integrated numerical model. Validated with measurements from Acoustic Doppler Current Profilers	[14]
Fiji	Marine current velocity	Measurements using current profilers and satellite data. Current recorded at intervals of 10 min averaged over 20 s. Also temperature and water level recorded	[138]
Iran Indonesia (Alas Street)	Surface velocity, flow power Current velocity	Current data of measurement stations, 5-min intervals, interpolated to other points Princeton Ocean Model (POM) used to create a numerical model of the Alas Strait. POM is a 3D, time dependent, sigma-coordinate, free surface model. Values of tidal ampli- tude, tidal phase and angular speed for each constituent and each node on the boundary, were obtained using the Oregon State University Ocean Tidal Prediction Software (OSU OTPS). Validated with measurements from Acoustic Doppler Current Profilers	[139] [140]
UK Southwest	Current velocity	2D finite element model. Turbine arrays modelled by means of a line sink of momentum	[26]
Canada, British Columbia	Tidal volume flux, extractable power	2D finite element model calculates tidal heights and currents	[9,141]
Norway	Kinetic energy flux	Current velocities gathered from DNL (mean maximum spring speed). Kinetic energy flux calculated from velocity and cross-section	[142]
Canada, Bay of Fundy	Tidal amplitude, tidal phase, theoretical turbine power	2D finite-volume model used (FVCOM). Model was run by specifying M2 phase and amplitude at open boundary	[143]
Taiwan, Kinmen Island	Current velocity, flow power density	3D semi-implicit, Euler-Lagrange finite-element model (SELFE) for simulation. Bottom topography data in the coastal seas from National Science Council, Taiwan. Validated with measurements from Acoustic Doppler Current Profilers and tide level measurements	[144]
UK, Pentland Firth	Water level amplitude, currents, extrac- table power	Depth-averaged numerical model. Actuator disc theory to model the effect of turbines on the flow, and to estimate the power available for generation after accounting for	[27]
US, South Carolina	Water level, mean flow velocity, max- imum flow velocity, power density	losses owing to mixing downstream of the turbines Regional Ocean Modelling System (ROMS) used. 3D free-surface, terrain-following, numerical model. ADCIRC tidal database used to define tidal constituents on seaward boundary. Field measurements used to validate results. Model used to simulate impact of the discipation of power due to the placement of extraction devices	[145]
China, Jiangsu	Flow velocity, power density	2D model built based on MIKE 21 to simulate tidal hydrodynamics. Time-dependent water elevation is provided along open-sea boundary. Validation of water elevation and current velocity with field mesurements.	[146]
Europe	Water level, current velocity, power density	European Tidal Database based on the Princeton Ocean Model (POM). Tide levels are forced at the model boundary with tidal components from the NAO99 dataset	[147]

forecasts for the sea computed by the operational circulation model cover 48 hours.

A number of commercial offers are available, mainly aiming at navigation (e.g. MaxSea [31]). They are mainly based on data available from public institutions such as NOAA or Meteorological Services.

2.3. Future research

According to [9], "further R&D is needed in the field of resource assessment, both on the measurement side (...), and on the development and validation of suitable modelling systems." However, in the recent years, modelling approaches have become more sophisticated and have been performed for many regions of the world. What is still missing is a harmonisation of approaches. In Europe, the SI Ocean project aims developing a "harmonized and comprehensive pan-European wave and tidal power resource map". Current activities of IEC TC 114 include the definitions of resource assessment requirement which will help in achieving harmonisation [32]. Going beyond pure resource assessment, it is necessary to connect resource assessment with local limitations from other activities in the marine environmental such as fishing, shipping, and offshore wind. Also other constraints are being included in resource assessments, for example protected habitats (Section 3.2), and the possibilities of grid connection (Section 5.2).

3. Environmental impacts

Ocean energy – as all other renewable sources of energy – can contribute to a more sustainable energy supply but it is not environmentally friendly per se. The activities involved in manufacturing, operation, maintenance and decommissioning of ocean energy devices will have various effects on the environment. Governments and society need a robust understanding of the environmental implications of ocean energy systems before ocean energy deployment takes place, and also to mitigate or adjust impacts to acceptable levels. While Environmental Impact Assessments (EIA) are performed to ensure that environmental implications of decisions are taken, Life Cycle Assessments (LCA) are used to identify and quantify the impact of industrial products on the environment.

The main direct expected environmental impacts of ocean wave and tidal current technology include impact on the benthic community (due to alterations in flow patterns, wave structures, sediment dynamics), species-specific response to habitat change, and the entanglement of marine mammals, turtles, larger fish and seabirds [33]. However, due to limited observations, the significance of environmental impacts of commercial deployment projects cannot fully be determined yet (see Section 3.2.1 on tidal currents and Section 3.2.2 on ocean wave). Future research in the area of environmental impacts should be focused on localised environmental impacts including e.g. electromagnetic field effects of subsea cables, flow alteration, sedimentation and habitat change of near generation devices. Examples of such efforts include [34,35] that model the impact on beach morphology exerted by wave energy farms or . Furthermore, it was stated that "comprehensive assessment, including both impacts and costs should be performed, applying the well-known Life Cycle Assessment (LCA) methodology to ocean energy generation" [9]. A new range of technologies, devices and sub-systems need in-depth analysis [9]. In addition, competing pressures and uses, e.g. climate change, fishing,

10

Results	Refer- ence	
Scores better in GHG, bulk waste, dangerous chemicals compared to national grid.	[148, 149]	
Energy and carbon intensities comparable to large wind turbines and very low relative to fossil-fuelled generation.	[150]	
curussons about 2.5 CO2/KWH. Based on [150] but CO2 expanded to GHG and more impact categories. GHG emissions about 30 g CO ₂ /kWh. Performs well in comparison to other removable encrines and foscil fiels.	[151]	
weith in comparison to out a terr weith entrance of the most rects. CO2 and energy demand calculated including operation and maintenance. Emissions about 25 g CO ₂ /RWh. Carbon intensity comparable to large wind turbines.	[152]	А.
Environmental impact of wave power stems mainly from construction, in particular, production of steel parts. Large possibilities to improve the environmental performance. GWP between 32 and 152 g/kWh.	[40]	Uihlein,
Carbon intensity significantly lower than PV and slightly higher than offshore wind. Much better as fossil.	[153]	D. Mag
Assumptions based on Crest project. Tidal about 2 g CO $_2$ /kWh. Better than wind, hydro and geothermal.	[41]	agna /
GHG emissions from wave and tidal less than 23 g CO2 eq/kWh, with a median estimate of 8 g CO2 eq/kWh for wave energy. In comparison to fossil energy, GHG emissions from ocean energy devices appear low	[2]	Renewable a
w fa n t fa e s fa fa e s fa fa c c t t t r a fa c c t t t r a fa c c c t t t r a fa e s t fa fa e s t fa fa e s t fa fa e s t fa e s t fa e s t fa fa fa fa fa fa fa fa fa fa fa fa fa	n n 2	nd Sustainal
	0 0	b

Energy CO₂ Energy CO₂

Cradle-to-Cradle-to-

grave grave

Hypothetical TEC array

Seagen Turbine

idal current

GHG

Cradle-to-

Range of literature

Wave and tidal

grave

GHG ODP AP POCP EP

Hypothetical WEC array

Oyster 1

Energy CO₂

Energy CO₂

Cradle-to-Cradle-to-Cradle-to-Cradle-to-

grave

grave grave grave grave

EDIP

Cradle-to

Wave Dragon Pelamis P1 Pelamis P1

Ocean wave

EDIP 2003

impact categories

Scope

FU (kWh)

narine transport should be considered when looking at environnental impacts [9].

.1. Life Cycle Assessment

The literature research shows that a small number of LCA on vave and tidal energy converters have been performed. The main ocus was on devices at already more advanced stage of developnent (e.g. Pelamis, Oyster, Seagen, and Wave Dragon). So far, most of ne studies addressed only energy and carbon as impact categories [2] concludes that there is a lack of studies of good quality. specially for tidal current. OTEC and salinity gradient devices. It is tated that "further LCA studies to increase the number of estimates or all ocean energy technologies are needed" [2].

Table 3 gives an overview of LCA studies on ocean energy ound. Several studies have not been included due to a lack of uality. For example, estimates of life cycle carbon emissions and nergy payback were given but device type, scope and boundaries f the assessment were not specified in [37]. Other studies inclued only the production of steel in the LCA for a wave energy onverter and only looked at selected emissions [38,39].

All studies performed so far focused on the amount of elecicity produced and did not look into issues like fluctuation, stoage, or grid integration. Arrays have not been considered except or the studies of [40,41]. For a number of WEC types like point bsorbers and marine current turbines, LCAs have not been ompiled yet.

2. Environmental impact assessment and strategic environmental assessment

Until now, there are still gaps concerning the scientific evidence on the environmental effects of ocean energy technologies [2,33]. Existing data are very much dispersed amongst countries, researchers and developers [42,43]. Since wave energy and tidal energy technologies are at an early development stage, no data on environmental effects from arrays are available.

Environmental impact assessments (EIA) and strategic environmental assessments (SEA) have been undertaken so far, with a focus on Europe and North America. Reviews on the state of research have been published by several institutions. The IEA-OES has summarised available knowledge on environmental impacts of ocean energy devices in three areas: physical interactions between animals and tidal turbines; acoustic impact on marine animals; and effects of energy removal on physical systems [44,45]. Similarly, the SOWFIA project aimed at sharing and consolidating experience of consenting processes and environmental and socioeconomic impact assessment best practices for wave energy [46].

3.2.1. Tidal current

In general, benthic habitats will be affected by tidal current energy converters and arrays due to the change of water flows, composition of substrate and sediment dynamics [33]. Potential other effects include mortality of fish passing through turbines (blade-strike) and the collision risk of marine mammals with tidal stream farms [33,47]. A study showed that change in sediment dynamics will most likely be observed following the installation of tidal arrays, impacting on bed morphology and benthic ecosystems [48]. This, in turn, could impact on floral and faunal species. Species of marine mammals and fish could experience distress and discomfort.

However, in their review, Frid et al. concludes that "there is little scientific literature to suggest that operation of underwater tidal stream energy devices will cause elevated levels of mortality to pelagic organisms such as fish and marine mammals" [33]. Also Lewis et al. mention that, "while current technologies have

Table 3

Overview of LCA studies on ocean energy

Device

Type

moving parts (rotating rotor blades or flapping hydrofoils) that may harm marine life, there is no evidence to date of harm from tidal current devices to marine life, such as whales, dolphins, seals and sharks" [2]. A critical issue related to tidal energy converters relates to the noise disruption in turbulent waters, affecting in particular marine mammals, who may be severely affected by such instance [49].

3.2.2. Ocean wave

For some devices, an Environmental Impact Assessments has been carried out, including AquaBuoy and Wave Dragon [50]. Wave energy converters can potentially "alter water column and sea bed habitats locally and by changes in the wave environment" [33]. A modelling exercise showed that the installation of wave energy converter arrays can lead to significant changes in the inter-array and surrounding wave field [51].

According to Lewis et al., environmental impact from ocean wave energy devices might include "competition for space, noise and vibration, electromagnetic fields, disruption to biota and habitats, water quality changes and possible pollution" [2]. As for tidal devices, the environmental impacts are considered comparably small [52,53]. Wave devices will represent a much lower collision risk compared to offshore wind devices but they could be the risk of underwater collisions for diving birds [54].

3.2. Future research

According to Lewis et al., "information on the environmental and social impacts is limited mainly due to the lack of experience in deploying and operating ocean technologies, although adverse environment effects are foreseen to be relatively low" [2]. In general, environmental impacts will very much depend on size of installation and the location selected [50]. Potential positive effects such as the creation of roosting sites and habitat enhancement for marine birds might occur as well [54].

The majority of the studies recommend that the first commercial scale installations of ocean energy technology should be accompanied by research studies on the local environmental impacts and for most installations, this will be covered by the EIA that is legally required.

Comprehensive LCA of ocean energy arrays are missing and an integration of aspects such as fluctuation of power output, storage, or grid integration would be very helpful. LCA of a number of major wave energy device types are still missing.

4. Socio-economic impacts

Socio-economic impact assessment addresses how a proposed development might affect the society as a whole or the local population. Various issues can be addressed ranging from wellbeing and quality of life to employment, income and economic power. For ocean energy, specific topics are negative effects due to visual impacts and the reduction of access to space for other users of the marine environment. Similar to the assessment of environmental impacts (Section 3), both positive and negative impacts of ocean energy deployment on society and economy needs to be studied in order to support evidence-based policy making.

Tools that are commonly used to assess socio-economic impacts are, for example, cost-benefit analysis or social impact assessment. Regional impacts or impacts on the whole economy can be assessed applying models such as computable general equilibrium models or empirical macro-economic models which also allow quantifying e.g. impacts on individual sectors, GDP, public budget, and household income. The FP7 project EquiMar aimed to deliver a suite of protocols for the equitable evaluation of marine energy converters (based on either tidal or wave energy) including economic assessment and cost modelling [55–57].

4.1. Costs

Many studies and reports have tried to estimate current and future costs of ocean energy. In general, there is a lack of data due to limited experience at commercial scale. The IEA-OES has published estimates for CAPEX and OPEX depending on the size of deployment (5–50 MW) and they have also addressed cost reduction pathways [42]. The SI-Ocean project has produced a report on cost of ocean energy in 2013 providing costs for early arrays and predictions for future costs by applying learning rates which was followed recently by a study from IEA-OES on the levelised cost of energy of ocean energy technologies [58,59]. Several recent journal articles have addressed specific issues by focussing on operation cost, costs for grid connection, or installation costs. Some examples are highlighted below.

An economic assessment of ocean energy has been performed by [60]. They highlight the lack of operational experience which means that operational costs have to be estimated. They provide a simulation model for operational costs and device availability to overcome these challenges. For the grid connection of an ocean energy array, Lopez et al. provide a preliminary cost estimate including a comparison of AV and DC transmission [61]. Cost components considered are offshore substation costs, cable costs, maintenance costs, and the costs for energy losses.

Morandeau et al. have developed a software tool for analysis and optimisation of marine energy installation [62]. The software incorporates metocean data, project information and vessel characteristics. It has been applied to a case study for two different types of installation vessels. An economic probabilistic model for marine offshore installations calculating CAPEX and OPEX and variation in LCoE of a hypothetical 250 MW wave array was developed by [51]. Their model captures effects of farm layout on energy productivity which allows optimisation of device siting.

An uncertainty analysis of financial aspects of wave energy farms was provided by [63]. The impacts of varying climate and wave conditions as well as feed-in tariffs on financial indicators such as IRR and NPV were quantified using a Monte Carlo method.

4.2. Social impacts

A main social impact that is usually addressed in studies on ocean energy is job creation. On national, European and global level, several estimates on the future potential for employment in the sector are available [64,65].

Other social impacts that have been addressed so far include CO_2 reductions [64], positive as well as negative impacts on other marine users [66] and local communities [67]; also the coexistence of fisheries and offshore renewable energy in the UK has been investigated [68]. It seems that other effects are difficult to quantify, including improvements to existing infrastructure; increased knowledge as a result of research and development in wave and tidal, improvements to energy security, health and quality of life.

4.3. Cost-benefit analysis

Only one cost–benefit analysis of a hypothetical tidal energy array in Nova Scotia was found in literature [69]. The case study was done for a hypothetical 300 MW farm and included several cost items (e.g. construction costs, operation and maintenance costs) and benefits (e.g. fuel savings, air quality, CO_2 emission reduction). In total, the net present value of the project was estimated to be negative due to the high initial capital costs. Unfortunately, no other cost–benefit analysis could be found to compare the results and findings, an especially to cover future installations including lower CAPEX due to learning curves.

4.4. Future research

Many studies address economic aspects of ocean energy including predictions of future costs of ocean energy. Still, there are improvements needed, especially in terms of operation and maintenance costs. Clearly, cost-benefit analyses including also aspects such as grid integration, energy security, and ecosystem services are missing. Social impacts are not well understood, when it comes to impacts beyond job creation such as welfare [70]. In particular, studies on the effects on national and EU level as well as coastal communities are needed [67,71].

5. Grid integration

The integration of ocean energy in the electricity grid poses various challenges. Firstly, ocean energy arrays have to be connected to the grid which can be very expensive since several components such as an array and subsea electrical system and a submarine cable connection to the shore are needed. Of course, grid availability in proximity to ocean energy arrays is a prerequisite for future developments [4]. Often, however, areas that offer good ocean energy resources are remote and often not connected with existing grid installation, thus requiring either grid upgrades or new-built capacity [5]. Grid integration is thus first of all an issue of electricity distribution and transmission and has also to be seen in the context of renewables integration at large [72].

Secondly, variability of electricity production from ocean energy devices might lead to issues such as grid congestion, weak grids, and voltage stability problems which is also related to technological development. Possible solutions and strategies to solve these problems are currently being researched and will be presented in the following sections.

5.1. Variability of resources

Ocean energy is a variable resource. Tidal current energy is periodic, and thus, resource forecasts are possible with a high reliability over long time horizons. Ocean wave energy can be considered a "stochastic" resource like wind energy [73]. For tidal currents, variability is very high on an hourly basis but limited for longer time horizons (e.g. monthly, yearly variation). On the contrary, variability for ocean wave energy is relatively low for short time scales (hours) but can be great for longer periods of time , for example on a seasonal or annual basis [2].

Reikard states that the forecast for wave energy is more precise than forecasts for wind and solar energy [20]. Still, the grid integration of ocean energy is of course influenced by the variability of resource availability. Several measures exist to accommodate for resource variability according to [74], including resource forecasting, intra- and inter-site smoothing, generation and load mix, and storage (e.g. pumped hydro, battery storage).

5.2. Grid connection and grid codes

In many cases, areas with great potential for ocean energy are located at regions with low population density with weak electricity networks. This may lead to a limitation of the electricity delivered to the grid due to quality of supply [75,76]. A reinforcement of transmission and distribution networks will most probably be necessary which implies high additional costs. Grid codes for transmission and distribution networks specify parameters such as frequency stability, voltage, power factor, and harmonics that an electricity generating facility has to meet in order to guarantee the safe operation of the electricity grid. Usually, distribution system operators establish the local regulations in distribution codes while transmissions system operators define the main transmission grid regulations.

Currently, no European standard grid code exist but the various national system operators issue their respective grid codes and usually, the requirements differ between countries [77]. At the moment, a range of countries are updating grid codes or developing new grid codes dedicate to the accommodation of a growing share of renewable electricity [78]. The MARINET project, funded by FP7 of the European Union produced a report on grid integration and power quality testing and reviewed the national grid codes [79,80]. The project highlighted that employing state-of-the-art technology from wind energy, such as frequency converters, will likely allow for grid-compliant installations of ocean energy farms [80].

5.3. Power quality and control

The power quality of ocean energy converters has to comply with distribution and transmission grid codes. MacEnri et al. checked performance of the SeaGen device against EN 50160 [82] in their study [81]. It concluded that "SeaGen works very well from a power quality perspective and is fully compliant with EN50160" [81]. It was considered unlikely that power quality problems will occur when arrays of the device will be installed.

In the future, renewable energy producers will face increasing demands on power quality to contribute to system reliability and stability. A number of requirements will have to be met including e.g. voltage control and regulation, fault ride-through capabilities, active power control, and frequency regulation [78]. IEC TC 114 is working on defining the electrical power quality requirements for wave, tidal and other water current energy converters in working group PT 62600–30 which will increase harmonisation amongst developers [32,83].

Demands on the control ocean energy converters and arrays will be of high importance [76]. This includes voltage and power factor control as well as power conditioning [75]. Hong et al. present an overview of control strategies for ocean energy devices including oscillating water column, attenuators, and overtopping devices [84]. They argue that further research on control strategies are needed since they also offer the "potential to dramatically affect the absorbed energy and hence the economy of the devices", which is also highlighted by [85]. Also Bacelli and Ringwood studied available control strategies for arrays of wave energy converters with respect to maximisation of energy absorption and conclude that performance can be increased by applying optimised control strategies [86]. The DTOcean project has a work package to identify, adapt and develop methods to optimise operational aspects of arrays of wave and tidal devices in terms of system control and operation [87].

Ongoing research tries to model impacts of ocean energy arrays on the grid by means of power flows and dynamic simulations. For example, Armstrong et al. studied the connection of hypothetical wave farms (one OWC farm and one heaving buoy farm) to the Western Interconnection (WECC) system [88]. Impacts on the transmission system were modelled in terms of voltage violations, loading conditions, congestion level, and substation performance. Impacts at the point of connection were quantified by assessing e.g. voltage level, flicker, harmonics, and low voltage ride through. It was shown that the integration of the wave energy farm would not pose any significant problems to the grid. Tedeschi and Santos-Mugica simulated the impacts of a wave farm (multi-MW point absorber) using the Spanish offshore testing facility Bimep as a real test case and studied different control options [89]. The model showed the importance of both wave and grid side energy wave energy converter control. In addition, centralised real-time control of the whole wave energy array reduced power variability and in consequence impacts on the grid.

5.4. Future research

Grid connection of ocean energy faces challenges due to weak electricity grids in rural areas. Reinforcement of grids will be needed which comes at high costs. The synergies with offshore wind farms need to be explored. At the moment, no single European standard grid code exists. Instead, the ocean energy sector – as other electricity producers – has to comply with the respective national or regional grid code. We will see increasing demands on power quality of renewable energy to contribute to system reliability and stability (e.g. voltage control and regulation, fault ride-through capabilities, active power control, and frequency regulation).

Most probably, the quality of power output from ocean energy arrays will meet the grid code requirements. However, adequate control systems for ocean energy converters and arrays have to be developed. Further research in the area of control strategies is needed since it offers a great potential for cost reduction due to increased absorbed energy while allowing meeting grid codes requirements.

6. Installation, operation and maintenance

The installation, operation and maintenance of ocean energy devices are relatively expensive. It is estimated that annual operation and maintenance costs of ocean energy devices can be as high as about 3.4–5.8% of capital expenditure compared to 2.3–3.7% for offshore wind [90]. One way to reduce those costs is to standardise equipment and procedures by industrial cooperation [4,6,91]. Other promising improvement options include the use of modelling tools to improve array layout and design which will lead to increased device and array efficiency and a reduction of costs.

6.1. Array configuration

Only few research papers have addressed issues of array configuration. One line of research looks at the impacts from ocean energy arrays on the hydrodynamic characteristics of the marine environment. One example of such a study is [48] who studied the effects of the configuration of a tidal array (turbine spacing and capacity) on water flows and water levels in the Shannon Estuary by means of a numerical model. They conclude that water flows and water levels will be affected, including a reduction in tidal range and a delay in high and low tides. Fallon et al. also develops suggestions for the optimisation of array spacing and location [48]. The FP7 project DTOcean ("Optimal Design Tools for Ocean Energy Arrays") has a dedicated work package on array layout. The project has already performed an assessment of capabilities of currently available tools studying the hydrodynamic interaction between the devices within the array but also with the resource and it will also determine how this affects the resource, power performance, cost uncertainties and environmental impact for selected scenarios [92]. It has to be noted, however, that array configuration will be highly dependent on the ocean energy device or technology chosen and flexibility will also be limited by the specific location chosen.

Going beyond impacts on hydrodynamics, some authors have tried to assess the influence of array configuration on power output as well. As an example [93], studied the impact of the shape and density of tidal current arrays $(1000 \times 10 \text{ m} \text{ diameter turbines})$ on hydro-environmental parameters and the energy output in the Severn Estuary and Bristol Channel. They concluded that the layout of a turbine array can have significant impacts on power output and, to a lesser extent, on hydro-environmental parameters. Lee et al. [94] modelled ocean current flows to optimise tidal current turbine array layout. They found that turbine efficiency increased with distance between generators and a distance of three times the turbine diameter was appropriate. Similarly, for wave energy, studies shave addressed modelling and optimisation of power output of arrays [95–97].

6.2. Installation

So far, only a few full-scale devices have been installed and thus practical experience is limited [75]. However, the ocean energy sector can build on technology and know-how from other offshore energy technologies [69,75]. Installation equipment from the oil and gas industry might be used but it could be too expensive since for ocean energy projects, the installation costs are responsible for a high share of investment costs [62,98]. Installation of ocean energy devices has to be easy and fast in order to reduce costs for the installation process [69]. In the case of tidal devices, this is also an important prerequisite because installation has to be performed during slack tide which means a limited time period.

The installation process and costs for wave and tidal devices will significantly depend on the location. For example, shore based wave energy converters might need solid foundations and heavy infrastructure. The same is the case for bottom-mounted tidal devices which demand substantial foundations. The mooring of floating devices with drag-anchors seems to be a very economical solution while in some cases the sea-bed characteristics will demand other and more expensive mooring types such as pin piled moorings [98]. Only few papers have tried to establish models or tools to assess resource needs for installation requirements in terms of time and cost.

6.3. Operation and maintenance

Ocean energy devices will operate in harsh environments. Demands on survivability and reliability are high since the economic impacts due to failures can be significant. Maintenance costs for ocean energy devices will be high as for any other offshore technology and have a high share of lifetime costs [58].

The most common issues ocean energy devices will face are bio-fouling (moorings, floating or submerged parts of the device) and corrosion [75,99]. Research needs to develop special coatings that prevent bio-fouling and corrosion but also sealing materials and electric insulation materials for saline environments [75,100]. Developers aim at reducing maintenance intervals by creating very robust devices and designing devices for ease-of-maintenance.

Current research tries to model the reliability and possible failure rates of ocean energy devoices. For example, Thies et al. developed a methodology to simulate component reliability and failure rates under defined operational conditions [101]. Device testing in environments that can produce the same conditions as in real waters is a prerequisite for assessing device and component reliability [69]. Also array design parameters (e.g. device spacing) impacts on maintenance activities and costs and this is not very well understood so far.

An important aspect that has to be taken into account when designing ocean energy devices and developing ocean energy projects is that maintenance and repair activities can only be carried out in favourable weather conditions. Weather window analyses study the levels of access in terms of a number of weather characteristics (e.g. wave heights, wind speed). Inaccessibility of ocean energy devices for maintenance and repair might require other maintenance strategies such as onshore maintenance in order to ensure economic viability of projects [102].

6.4. Future research

Arrays of ocean energy converters have not been installed yet. However, models have been created that allow capturing the effects of arrays on hydrodynamics as well as power output. Since no long-term experience with devices is available concerning commercial operation and maintenance, there are only a few articles published that try to assess the resource needs for installation (e.g. time, cost). Array design parameters such as device spacing might have an impact on operation and maintenance activities and costs: this is not very well understood so far. Concerning reliability and performance, current research tries to model the reliability and possible failure rates of ocean energy devices.

7. Regulatory and legal affairs

Ocean energy is promoted by governments as a renewable source of energy that will help reaching climate targets and contribute to a secure energy supply. However, often not much attention is given to the existing legal framework and legal barriers may hinder the development of ocean energy [103,104]. It is thus important to study the legal framework to analyse the current legal situation, to identify best approaches and conflicting regulations [104,105].

7.1. Legal framework

The UN Convention on the Law of the Sea defines the main legal framework for the use of the oceans [106,107]. According to the Convention, territorial waters or territorial sea encompasses the coastal waters up to 12 nautical miles from a baseline which usually is the mean low-water mark. Territorial waters are sovereign territory of the state and give the full rights over water, sea bed and subsoil. The coastal country has the right to set laws, regulate the use of the ocean in its territorial waters [107]. The territorial waters are followed by an exclusive economic zone extending up to 200 nautical miles from the territorial sea baseline where a country possesses the rights to explore, exploit, conserve and manage the natural resources of the water column and seabed. Most ocean energy devices have and will be installed in territorial waters [107].

In many countries, there is no specific legal and regulatory framework for ocean energy in territorial waters [108]. For example, in the case of Ireland, existing environmental and maritime legislation applies to ocean energy including the following legislative instruments [103]:

- Foreshore acts (e.g. [109,110]);
- Electricity regulations (e.g. [111–113]);
- EU EIA Directive [114,115];
- EU Habitats and birds directive [116];
- EU SEA Directive [117];
- EU Renewable Energy Directive [118].

Legal and regulatory issues are a barrier towards ocean energy deployment. Leary and Esteban state that regulatory uncertainty appears to be the most significant non-technical barrier to the ocean energy sector [119]. They mention that in the USA, more than 23 federal and state regulatory agencies are involved in ocean energy projects. Often, approval processes are complex due to the large range of legislation and regulatory bodies. O'Hagan and Lewis mention Ireland as another example, where "management of all maritime sectors proceeds largely in an ad hoc, non- integrated manner with little formal inter-departmental communication" [103]. A comparison of national policy frameworks of the UK and France performed by the MERiFIC project showed that there are substantial differences between countries in terms of consenting procedures and "fragmentary evolution [...] of consenting procedures for marine renewables" leads to complications [120].

The H2020 project RiCORE funded from the EU is a research project studying the legal framework for offshore renewable energy in EU Member States. It will support the improvement of consenting procedures by providing best practices [121].

7.2. Maritime and spatial planning

Marine spatial planning is considered as a solution to overcome problems with overlapping jurisdiction and to support ocean energy to tap the full potential [103,122]. According to [123], it can "help to avoid user conflicts, to improve the management of marine spatial claims, and to sustain an ecosystem-based management of ocean and seas".

Maritime Spatial Planning (MSP) has not been fully introduced in all EU Member States so far [104,124]. In 2013, the European Commission has proposed a Directive on maritime spatial planning and integrated coastal management in 2013 which has been adopted by the European Parliament in April 2014 [125]. The Directive will make maritime spatial planning and integrated coastal management mandatory in the EU Member States. Maritime spatial planning is an important pillar in the Commissions' strategy for blue growth. It is hoped that over-regulation and administrative complexity can be reduced allowing for accelerated investment in the marine environment and generating economic gains [126,127].

Case studies using applying marine spatial planning for the ocean energy sector have already proven that it offers the potential of an integrated management of marine resource use [128,129]. In addition, it offers the potential to optimise siting location for ocean energy devices by accommodating the economic constraints for ocean energy production, conflicts with other marine users and for minimising environmental impacts.

7.3. Future research

The research in the area of regulatory and legal affairs that has been executed so far has focussed on consenting procedures and the approval process for ocean energy projects. In addition, the advantages of integrated management tools such as maritime spatial planning have been highlighted. Still, more detailed research is needed concerning regulations and legal frameworks [130]. Wright suggests comparing jurisdiction of some more advanced in order to identify the most optimal framework [131].

8. Conclusions

This study has reviewed the state-of-research in ocean energy, focusing on wave and tidal current, not directly associated with improvement to ocean energy technology and identified areas where future research efforts should be directed to.

Modelling approaches for resource assessment and forecasting are already very advanced and have been performed for many regions of the world. However, this should be widened to accommodate conflicting or competing use of the marine environment such as fishing, shipping, offshore wind, habitat protection and also technical limitations (e.g. grid connection). Comprehensive Life Cycle Assessments of ocean energy arrays that would also include areas like fluctuation of power output, storage, or grid integration are still missing and for a number of individual WEC types, no Life Cycle Assessments are available so far. Another area which merits further research is the field of regulatory and legal affairs to define an adequate and optimal legal framework for ocean energy.

In terms of grid integration, the impacts of increasing demands on power quality of renewable energy to contribute to system reliability and stability should be discussed. Further research in the area of control strategies is needed since it offers a great potential for cost reduction due to increased absorbed energy while allowing meeting grid codes requirements. No long-term experience with devices is available concerning commercial operation and maintenance and few articles try to assess the resource needs for installation (e.g. time, cost). Array design parameters such as device spacing might have an impact on operation and maintenance activities and costs: this is not very well understood so far and should be addressed.

The most important areas, however, where future research should be focussing on are the economic and social impacts of ocean energy. A broad cost benefit analysis of ocean energy incorporating aspects such as grid integration and energy security could be very important. Economic aspects of ocean energy including predictions of future costs of ocean energy have been addressed but improvements are needed, especially in the area of operation and maintenance costs. Cost-benefit analyses that include also aspects such as grid integration, energy security, and ecosystem services are missing. Social impacts are not well understood, when it comes to impacts beyond job creation. In particular, studies on the effects on national and EU level are needed.

9. Disclaimer

The views expressed in this paper are purely those of the writer and may not in any circumstances be regarded as stating an official position of the European Commission.

References

- Huckerby J, Jeffrey H, Jay B. An international vision for ocean energy. Ocean energy systems implementing agreement; 2011.
- [2] Lewis A, Estefen S, Huckerby J, Musial W, Pontes T, Torres-Martinez J. Ocean Energy. et al.. In: Edenhofer O, Pichs-Madruga R, Sokona Y, Seyboth K, Matschoss P, Kadner S, editors. IPCC special report on renewable energy sources and climate change mitigation. Cambridge (New York): Cambridge University Press; 2011. p. 497–533.
- [3] Brito E Melo A, Villate JL, editors. Annual report 2014. Implementing Agreement on ocean energy systems. IEA-OES; 2015. <u>http://dx.doi.org/10.</u> 1017/S000197200000176.
- [4] Magagna D, MacGillivray A, Jeffrey H, Hanmer C, Raventos A, Badcock-Broe A, et al. Wave and tidal energy strategic technology agenda. Strategic initiative for ocean energy (SI ocean); 2014.
- [5] Magagna D, Uihlein A. JRC ocean energy status report. Luxembourg: Publications Office of the European Union; <u>http://dx.doi.org/10.2790/866387</u>.
- [6] MacGillivray A, Jeffrey H, Hanmer C, Magagna D, Raventos A, Badcock-Broe A. Ocean energy technology: gaps and barriers. Strategic initiative for ocean energy (SI ocean); 2013.
- [7] EERA and ERA-NET joint workshop on wave and tidal energy within the European Union. European Energy Research Alliance; 2014.
- [8] Magagna D, Uihlein A. Ocean energy development in Europe: current status and future perspectives. Int J Mar Energy 2015;11:84–104. <u>http://dx.doi.org/</u> <u>10.1016/j.ijome.2015.05.001.</u>
- [9] Sannino G, Cavicchioli C. Overcoming research challenges for ocean renewable energy. Luxembourg: European Union; <u>http://dx.doi.org/10.2790/8776</u>.
- [10] Iglesias G, Carballo R. Choosing the site for the first wave farm in a region: a case study in the Galician Southwest (Spain). Energy 2011;36:5525–31. <u>http:</u> //dx.doi.org/10.1016/j.energy.2011.07.022.

- [11] Arinaga RA, Cheung KF. Atlas of global wave energy from 10 years of reanalysis and hindcast data. Renew Energy 2012;39:49–64. <u>http://dx.doi.org/</u> 10.1016/j.renene.2011.06.039.
- [12] Gonçalves M, Martinho P, Guedes Soares C. Wave energy conditions in the western French coast. Renew Energy 2014;62:155–63. <u>http://dx.doi.org/</u> 10.1016/j.renene.2013.06.028.
- [13] Carballo R, Iglesias G. A methodology to determine the power performance of wave energy converters at a particular coastal location. Energy Convers Manag 2012;61:8–18. http://dx.doi.org/10.1016/j.enconman.2012.03.008.
- [14] O'Rourke F, Boyle F, Reynolds A. Tidal current energy resource assessment in Ireland: Current status and future update. Renew Sustain Energy Rev 2010;14:3206–12. http://dx.doi.org/10.1016/j.rser.2010.07.039.
- [15] Gunn K, Stock-Williams C. Quantifying the global wave power resource. Renew Energy 2012;44:296–304. <u>http://dx.doi.org/10.1016/j.renene.2012.01.101</u>.
- [16] Strategic Research Agenda/Market Deployment Strategy. European wind energy technology platform; 2014.
- [17] Cornett AM. A global wave energy resource assessment. In: Proceedings of the eighteenth international offshore polar engineering conference. Vancouver: The International Society of Offshore and Polar Engineers (ISOPE); 2008, p. 318–27.
- [18] Akpinar A, Kömürcü Mİ. Assessment of wave energy resource of the Black Sea based on 15-year numerical hindcast data. Appl Energy 2013;101:502– 12. http://dx.doi.org/10.1016/j.apenergy.2012.06.005.
 [19] Reikard G, Pinson P, Bidlot J-R. Forecasting ocean wave energy: the ECMWF
- [19] Reikard G, Pinson P, Bidlot J-R. Forecasting ocean wave energy: the ECMWF wave model and time series methods. Ocean Eng 2011;38:1089–99. <u>http:</u> //dx.doi.org/10.1016/j.oceaneng.2011.04.009.
- [20] Reikard G. Integrating wave energy into the power grid: simulation and forecasting. Ocean Eng 2013;73:168–78. <u>http://dx.doi.org/10.1016/j.oceaneng.2013.08.005</u>.
- [21] Malmberg A, Holst U, Holst J. Forecasting near-surface ocean winds with Kalman filter techniques. Ocean Eng 2005;32:273–91. <u>http://dx.doi.org/</u> 10.1016/j.oceaneng.2004.08.005.
- [22] Ho PC, Yim JZ. Wave height forecasting by the transfer function model. Ocean Eng 2006;33:1230–48. <u>http://dx.doi.org/10.1016/j.oceaneng.2005.09.003</u>.
- [23] Londhe SN, Panchang V. One-day wave forecasts based on artificial neural networks. J Atmos Ocean Technol 2006;23:1593–603. <u>http://dx.doi.org/</u> 10.1175/JTECH1932.1.
- [24] Roulston MS, Ellepola J, Hardenberg J, von, Smith LA. Forecasting wave height probabilities with numerical weather prediction models. Ocean Eng 2005;32:1841–63. <u>http://dx.doi.org/10.1016/j.oceaneng.2004.11.012</u>.
- [25] Gaur S, Deo MC. Real-time wave forecasting using genetic programming. Ocean Eng 2008;35:1166–72. http://dx.doi.org/10.1016/j.oceaneng.2008.04.007.
- [26] Serhadlioğlu S, Adcock TAA, Houlsby GT, Draper S, Borthwick AGL. Tidal stream energy resource assessment of the Anglesey Skerries. Int J Mar Energy 2013. http://dx.doi.org/10.1016/j.ijome.2013.11.014.
- [27] Adcock TAA, Draper S, Houlsby GT, Borthwick AGL, Serhadlioglu S. The available power from tidal stream turbines in the Pentland Firth. Proc R Soc A Math Phys Eng Sci 2013. <u>http://dx.doi.org/10.1098/rspa.2013.0072</u>.
- [28] Lynn PA. Electricity from wave and tide: an introduction to marine energy. Chichester: Wiley; 2013.
- [29] Prediction models. (http://www.bsh.de/en/Marine_data/Forecasts/Predic tion_models/index.jsp); 2015 [accessed 07.07.15].
- [30] Weather and Climate Deutscher Wetterdienst Seewetter. (http:// www.dwd.de/bvbw/appmanager/bvbw/dwdwwwDesktop? _nfpb=true&_pageLabel=_dwdwww_spezielle_nutzer_schifffahrt_ seewetter&T18605718401151909352281gsbDocumentPath=Navigation %2FSchifffahrt%2FSeewetter%2FHome_NO_node.html%3F_nnn% 3Dtrue); 2015 [accessed 07.07.15].
- [31] High Resolution Tidal Currents. (http://www.maxsea.com/products/charts/ currents); 2015 [accessed 07.07.15].
- [32] IEC/TC 114 Strategic Business Plan. International Electrotechnical Commission (IEC); 2013.
- [33] Frid C, Andonegi E, Depestele J, Judd A, Rihan D, Rogers SI, et al. The environmental interactions of tidal and wave energy generation devices. Environ Impact Assess Rev 2012;32:133–9. <u>http://dx.doi.org/10.1016/j.</u> eiar.2011.06.002.
- [34] Abanades J, Greaves D, Iglesias G. Wave farm impact on beach modal state. Mar Geol 2015;361:126–35. http://dx.doi.org/10.1016/j.margeo.2015.01.008.
- [35] Abanades J, Greaves D, Iglesias G. Wave farm impact on the beach profile: a case study. Coast Eng 2014;86:36–44. <u>http://dx.doi.org/10.1016/j. coastaleng.2014.01.008.</u>
- [36] Raventós A, Simas T, Moura A, Harrison G, Thomson C, Dhedin J-F. Life cycle assessment for marine renewables; 2010.
- [37] Callaghan J, Boud R. Future marine energy results of the marine energy challenge: cost competitiveness and growth of wave and tidal stream energy. London: Carbon Trust; 2006.
- [38] Banerjee S, Duckers LJ, Blanchard R. Development of an assessment tool for wave energy systems. In: World renewable energy congress, Florence: 2006, p. 6.
- [39] Banerjee S, Duckers L, Blanchard RE. An overview on greenhouse gas emission characteristics and energy evaluation of ocean energy systems from life cycle assessment and energy accounting studies. J Appl Nat Sci 2013;5:535–40.
- [40] Dahlsten H. Life cycle assessment of electricity from wave power. Swedish University of Agricultural Sciences; 2009.
- [41] Rule BM, Worth ZJ, Boyle CA. Comparison of life cycle carbon dioxide emissions and embodied energy in four renewable electricity generation

technologies in New Zealand. Environ Sci Technol 2009;43:6406–13. <u>http:</u>//dx.doi.org/10.1021/es900125e.

- [42] Brito E Melo A, Huckerby J, editors. Annual report 2012. Implementing agreement on ocean energy systems. Lisboa: IEA-OES; 2012.
- [43] Brito E Melo A, Villate JL, editors. Annual report 2013. Implementing agreement on ocean energy systems. Lisboa: IEA-OES; 2013.
- [44] Copping A, Battey H, Brown-Saracino J, Massaua M, Smith C. An international assessment of the environmental effects of marine energy development. Ocean Coast Manag 2014;99:3–13. http://dx.doi.org/10.1016/j.ocecoaman.2014.04.002.
- [45] Copping A, Hanna L, Whiting J, Geerlofs S, Grear M, Coffey A, et al. Environmental effects of marine energy development around the world. Annex IV final report. IEA-OES: 2013.
- [46] Conley D, Magagna D, Greaves D, Aires E, Leitao JC, Witt M, et al. Work package 3 final report. Deliverable 3.5. 2013.
- [47] Boehlert G, Gill A. Environmental and ecological effects of ocean renewable energy development – a current synthesis. Oceanography 2010;23:68–81. http://dx.doi.org/10.5670/oceanog.2010.46.
- [48] Fallon D, Hartnett M, Olbert A, Nash S. The effects of array configuration on the hydro-environmental impacts of tidal turbines. Renew Energy 2014;64:10–25. <u>http://dx.doi.org/10.1016/j.renene.2013.10.035</u>.
- [49] Polagye B, Cleve B Van, Kirkendall K, Copping A. Environmental effects of tidal energy development. In: Proceedings of a scientific workshop. Seattle: National Oceanic and Atmospheric Administration (NOAA); 2011.
- [50] Margheritini L, Hansen AM, Frigaard P. A method for ElA scoping of wave energy converters—based on classification of the used technology. Environ Impact Assess Rev 2012;32:33–44. <u>http://dx.doi.org/10.1016/j.eiar.2011.02.003</u>.
- [51] Posner AJ, O' Sullivan K, Murphy J. Economic and environmental impact appraisal of commercial scale offshore renewable energy installations on the west coast of Ireland. J Coast Res 2013;29:1639–44. <u>http://dx.doi.org/</u> 10.2112/SI65-277.1.
- [52] Clément A, McCullen P, Falcão A, Fiorentino A, Gardner F, Hammarlund K, et al. Wave energy in Europe: current status and perspectives. Renew Sustain Energy Rev 2002;6:405–31. <u>http://dx.doi.org/10.1016/S1364-0321(02)</u> 00009-6.
- [53] Fadaeenejad M, Shamsipour R, Rokni SD, Gomes C. New approaches in harnessing wave energy: With special attention to small islands. Renew Sustain Energy Rev 2014;29:345–54. http://dx.doi.org/10.1016/j.rser.2013.08.077.
- [54] Grecian WJ, Inger R, Attrill MJ, Bearhop S, Godley BJ, Witt MJ, et al. Potential impacts of wave-powered marine renewable energy installations on marine birds. Ibis 2010;1859(152):683–97. <u>http://dx.doi.org/10.1111/</u> j.1474-919X.2010.01048.x.
- [55] Equitable testing and evaluation of marine energy extraction devices in terms of performance, cost and environmental impact; 2011.
- [56] Ricci P, Lopez J, Villate J, Stallard T. Summary of attributes of cost models used by different stakeholders. EquiMar - Deliverable 7.1. EquiMar; 2009.
- [57] Davey T, Harrison GP, Stallard T. Procedures for economic evaluation. Equi-Mar - Deliverable 7.2.1. EquiMar; 2009.
- [58] Ocean energy: cost of energy and cost reduction opportunities. Strategic initiative for ocean energy (SI ocean); 2013.
- [59] International levelised cost of energy for ocean energy technologies. Lisbon: IEA-OES; 2015.
- [60] Teillant B, Costello R, Weber J, Ringwood J. Productivity and economic assessment of wave energy projects through operational simulations. Renew Energy 2012;48:220–30. http://dx.doi.org/10.1016/j.renene.2012.05.001.
- [61] Lopez J, Ricci P, Villate JL, Bahaj AS, Myers LE, Retzler C, et al. Preliminary economic assessment and analysis of grid connection schemes for ocean energy arrays. In: Proceedings of the 3rd international conference on ocean energy, Bilbao; 2010.
- [62] Morandeau M, Walker RT, Argall R, Nicholls-Lee RF. Optimisation of marine energy installation operations. Int J Mar Energy 2013;3-4:14–26. <u>http://dx. doi.org/10.1016/j.ijome.2013.11.002.</u>
- [63] Guanche R, de Andrés AD, Simal PD, Vidal C, Losada IJ. Uncertainty analysis of wave energy farms financial indicators. Renew Energy 2014;68:570–80. http://dx.doi.org/10.1016/j.renene.2014.02.046.
- [64] Impact Assessment. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions: Ocean Energy. Action needed to deliver on the potential of ocean energy by 2020. Brussels: European Commission; 2014.
- [65] Industry Vision Paper 2013. Brussels: European Ocean Energy Association; 2013.
- [66] A socio-economic methodology and baseline for Pentland Firth and Orkney waters wave and tidal developments. The Crown Estate; 2012.
- [67] Henkel SK, Conway FDL, Boehlert GW. Environmental and human dimensions of ocean renewable energy development. Proc IEEE 2013;101:991–8. http://dx.doi.org/10.1109/JPROC.2013.2239598.
- [68] De Groot J, Campbell M, Ashley M, Rodwell L. Investigating the co-existence of fisheries and offshore renewable energy in the UK: identification of a mitigation agenda for fishing effort displacement. Ocean Coast Manag 2014;102:7–18. http://dx.doi.org/10.1016/j.ocecoaman.2014.08.013.
- [69] Houde J, Cost-Benefit. Analysis of tidal energy generation in Nova Scotia: a scenario for a tidal farm with 300 MW of installed capacity in the Minas Passage in 2020. Dalhousie University: 2012.
- [70] Deane JP, Dalton G, Ó Gallachóir BP. Modelling the economic impacts of 500 MW of wave power in Ireland. Energy Policy 2012;45:614–27. <u>http://dx. doi.org/10.1016/j.enpol.2012.03.012</u>.

- [71] Kerr S, Watts L, Colton J, Conway F, Hull A, Johnson K, et al. Establishing an agenda for social studies research in marine renewable energy. Energy Policy 2014;67:694–702. <u>http://dx.doi.org/10.1016/j.enpol.2013.11.063</u>.
- [72] ENTSOE. Ten-year network development plan 2014. (https://www.entsoe.eu/ major-projects/ten-year-network-development-plan/Pages/default.aspx); 2014 [accessed 11.11.14].
- [73] Iyer AS, Couch SJ, Harrison GP, Wallace AR. Variability and phasing of tidal current energy around the United Kingdom. Renew Energy 2013;51:343–57. http://dx.doi.org/10.1016/j.renene.2012.09.017.
- [74] Santos Múgica M, Salcedo Fernandez F, Ben Haim D, Lopez Mendia J, Ricci P, Villate Martínez JL, et al. Integrating wave and tidal current power: case studies through modelling and simulation. Tecnalia, Powertech Labs, HMRC; 2011.
- [75] Mueller M, Wallace R. Enabling science and technology for marine renewable energy. Energy Policy 2008;36:4376–82. <u>http://dx.doi.org/10.1016/j.</u> enpol.2008.09.035.
- [76] Kiprakis AE, Wallace AR. Maximising energy capture from distributed generators in weak networks. IEE Proc – Gener Transm Distrib 2004;151:611. http://dx.doi.org/10.1049/ip-gtd:20040697.
- [77] Lorente Iglesias R, Lacal Arantegui R, Aguado Alonso M. Power electronics evolution in wind turbines—a market-based analysis. Renew Sustain Energy Rev 2011;15:4982–93. <u>http://dx.doi.org/10.1016/j.rser.2011.07.056</u>.
- [78] Grid integration of large-capacity renewable energy sources and use of large-capacity electrical energy storage. Geneva: International Electrotechnical Commission; 2012.
- [79] Santos-Mugica M, Robles E, Endegnanew AG, Tedeschi E, Giebhardt J. Grid integration and power quality testing of marine energy converters: research activities in the MARINET project. In: Proceedings of the 2014 ninth international conference on ecological vehicles and renewable energies, Monte Carlo; 2014, p. 9.
- [80] Giebhardt J, Kracht P, Dick C, Salcedo F. Report on grid integration and power quality testing. Deliverable 4.3 final. MARINET; 2014.
- [81] MacEnri J, Reed M, Thiringer T. Power quality performance of the tidal energy converter, SeaGen. In: Proceedings of the ASME 2011 30th international conference on ocean offshore and arctic engineering, Rotterdam; 2011. p. 8.
- [82] EN 50160:2010. Voltage characteristics of electricity supplied by public electricity networks; 2010.
- [83] IEC TC 114 Work programme. Project: IEC/TS 62600-30 2015. (http://www.iec. ch/dyn/www/f?p=103:38:0::::FSP_ORG_ID,FSP_APEX_PAGE,FSP_LANG_ID,FSP_ PROJECT:1316,23,25,IEC/TS 62600-30 Ed. 1.0#); 2015 [accessed 07.07.15].
- [84] Hong Y, Waters R, Boström C, Eriksson M, Engström J, Leijon M. Review on electrical control strategies for wave energy converting systems. Renew Sustain Energy Rev 2014;31:329–42. <u>http://dx.doi.org/10.1016/j.rser.2013.11.053</u>.
- [85] Sanchez EV, Hansen RH, Kramer MM. Control performance assessment and design of optimal control to harvest ocean energy. IEEE J Ocean Eng 2014:1– 12. http://dx.doi.org/10.1109/JOE.2013.2294386.
- [86] Bacelli G, Ringwood J. Constrained control of arrays of wave energy devices. Int J Mar Energy 2013;3–4:e53–69. <u>http://dx.doi.org/10.1016/j.ijome.2013.11.011</u>.
- [87] System Control & Operation. (http://www.dtocean.eu/Work-Packages/System-Control-Operation); n.d. [accessed 09.07.15].
- [88] Armstrong S, Cotilla-Sanchez E, Kovaltchouk T. Assessing the impact of the grid connected Pacific marine energy centre wave energy farm. IEEE J Emerg Sel Top Power Electron 2015. <u>http://dx.doi.org/10.1109/JESTPE.2015.2429577</u>.
- [89] Tedeschi E, Santos-Mugica M. Modeling and control of a wave energy farm including energy storage for power quality enhancement: the bimep case study. IEEE Trans Power Syst 2014;29:1489–97. <u>http://dx.doi.org/10.1109/</u> <u>TPWRS.2013.2282213</u>.
- [90] Lacal-Arantegui R, Jäger-Waldau A, Vellei M, Sigfusson B, Magagna D, Jakubcionis M, et al. DRAFT 2014 energy technology reference indicator (ETRI) projections for 2010–2050. Luxembourg: Publications Office of the European Union; 2014.
- [91] Badcock-Broe A, Flynn R, George S, Gruet R, Medic N. Wave and tidal energy market deployment strategy for Europe. Strategic initiative for ocean energy (SI ocean); 2014.
- [92] Têtu A, Kofoed JP, Tully S, Roc T. Assessment of capabilities of available tools. DTOcean – Deliverable 2.1. DTOcean; 2014.
- [93] Ahmadian R, Falconer RA. Assessment of array shape of tidal stream turbines on hydro-environmental impacts and power output. Renew Energy 2012;44:318–27. http://dx.doi.org/10.1016/j.renene.2012.01.106.
 [94] Lee SH, Lee SH, Jang K, Lee J, Hur N. A numerical study for the optimal
- [94] Lee SH, Lee SH, Jang K, Lee J, Hur N. A numerical study for the optimal arrangement of ocean current turbine generators in the ocean current power parks. Curr Appl Phys 2010;10:S137–41. <u>http://dx.doi.org/10.1016/j.</u> <u>cap.2009.11.018</u>.
- [95] Wolgamot HA, Taylor PH, Eatock Taylor R. The interaction factor and directionality in wave energy arrays. Ocean Eng 2012;47:65–73. <u>http://dx.doi.org/</u> 10.1016/j.oceaneng.2012.03.017.
- [96] Child BFM, Venugopal V. Optimal configurations of wave energy device arrays. Ocean Eng 2010;37:1402–17. <u>http://dx.doi.org/10.1016/j.oceaneng.2010.06.010</u>.
- [97] Borgarino B, Babarit A, Ferrant P. Impact of wave interactions effects on energy absorption in large arrays of wave energy converters. Ocean Eng 2012;41:79–88. <u>http://dx.doi.org/10.1016/j.oceaneng.2011.12.025</u>.
- [98] Ocean energy: state of the art. Strategic initiative for ocean energy (SI ocean); 2012.
- [99] Walker JM, Flack KA, Lust EE, Schultz MP, Luznik L. Experimental and numerical studies of blade roughness and fouling on marine current turbine

performance. Renew Energy 2014;66:257–67. <u>http://dx.doi.org/10.1016/j.</u>renene.2013.12.012.

- [100] Boisseau A, Davies P, Thiebaud F. Sea water ageing of composites for ocean energy conversion systems: Influence of glass fibre type on static behaviour. Appl Compos Mater 2012;19:459–73. <u>http://dx.doi.org/10.1007/</u> s10443-011-9219-6.
- [101] Thies PR, Johanning L, Smith GH. Towards component reliability testing for marine energy converters. Ocean Eng 2011;38:360–70. <u>http://dx.doi.org/</u> 10.1016/j.oceaneng.2010.11.011.
- [102] O'Connor M, Lewis T, Dalton G. Weather window analysis of Irish and Portuguese wave data with relevance to operations and maintenance of marine renewables. In: Proceedings of the ASME 2013 32nd international conference on ocean offshore and arctic engineering, vol. 8. Ocean renewable energy. ASME; 2013. http://dx.doi.org/10.1115/OMAE2013-11125.
- [103] O'Hagan AM, Lewis AW. The existing law and policy framework for ocean energy development in Ireland. Mar Policy 2011;35:772–83. <u>http://dx.doi.org/10.1016/j.marpol.2011.01.004</u>.
- [104] Simas T, O'Hagan A-M, Bailey I, Marina D, Sundberg J, Le Crom I, et al. Consenting procedures review with guidelines for expansion to larger projects and approval process streamlining, incorporating the findings of interim report and feedback from workshop D. Deliverable D.4.6. Final work package report; 2013.
- [105] Kannen A, Kremer H, Gee K, Lange M. Renewable energy and marine spatial planning: scientific and legal implications. In: Nordquist MH, Moore JN, Chircop A, Long R, editors. Regulation of continental shelf development, Brill; 2013, p. 151–78. http://dx.doi.org/10.1163/9789004256842_009.
- [106] United Nations Convention on the Law of the Sea of 10 December 1982, vol. UNCLOS; 1982.
- [107] Abad Castelos M. Marine renewable energies: opportunities, law, and management. Ocean Dev Int Law 2014;45:221–37. <u>http://dx.doi.org/10.1080/</u> 00908320.2014.898926.
- [108] Wright G. Ocean energy: a legal perspective. J Ocean Technol 2013;8:26–32.
 [109] Foreshore (amendment) Act 2011. (http://www.irishstatutebook.ie/pdf/2011/ en.act.2011.0011.pdf); n.d. [accessed 02.10.14].
- [110] Ley 2/2013, de 29 de mayo, de protección y uso sostenible del litoral y de modificación de la Ley 22/1988, de 28 de julio, de Costas; n.d.
- [111] Energy Policy Act of 2005; 2005.
- [112] Gesetz für den Ausbau erneuerbarer Energien; 2014.
- [113] Electricity Act 1989; 1989.
- [114] Council Directive 85/337/EEC of 27 June 1985 on the assessment of the effects of certain public and private projects on the environment; 1985.
- [115] Directive 2011/92/EU of the European Parliament and of the Council of 13 December 2011 on the assessment of the effects of certain public and private projects on the environment; 2011.
- [116] Council Directive 92/43/EEC of 21 May 1992 on the conservation of natural habitats and of wild fauna and flora; 1992.
- [117] Directive 2001/42/EC of the European Parliament and of the Council of 27 June 2001 on the assessment of the effects of certain plans and programmes on the environment; 2001.
- [118] Directive 2009/28 of the European Parliament and of the Council of 23 April 2009 on the promotion of the use of energy from renewable sources and amending and subsequently repealing Directives 2001/77/EC and 2003/30/ EC/EC; 2009.
- [119] Leary D, Esteban M. Climate change and renewable energy from the ocean and tides: calming the sea of regulatory uncertainty. Int J Mar Coast Law 2009;24:617–51. http://dx.doi.org/10.1163/092735209X12499043518269.
- [120] Bailey I, Groot J de, Whitehead I, Vantoch-Wood A, Peter Connor. Comparison of national policy frameworks for marine renewable energy within the United Kingdom and France. MERiFIC – Task 4.1.2. MERiFIC; 2012.
- [121] About RiCORE 2015. (http://ricore-project.eu/); 2015 [accessed 09.07.15].
- [122] Righi A. Rough seas for renewable energy: addressing regulatory overlap for hydrokinetic projects on the outer continental shelf. Washingt J Environ Law Policy 2011;1:79–128.
- [123] Maes F. The international legal framework for marine spatial planning. Mar Policy 2008;32:797–810. http://dx.doi.org/10.1016/j.marpol.2008.03.013.
- [124] Barten S. EU Member States remain in charge of maritime spatial planning: the new maritime spatial planning directive; 2014. (http://www.main portlawyers.com/content/eu-member-states-remain-charge-maritime-spa tial-planning-new-maritime-spatial-planning).
- [125] Proposal for a Directive of the European Parliament and of the Council establishing a framework for maritime spatial planning and integrated coastal management. vol. COM(2013); 2013.
- [126] Commission welcomes Parliament's adoption of maritime spatial planning legislation. Brussels; 2014.
- [127] Vantoch-Wood A, de Groot J, Connor P, Bailey I, Whitehead I. National policy framework for marine renewable energy within the United Kingdom. MERiFIC Task 4.1.1. MERiFIC; 2012.
- [128] Galparsoro I, Liria P, Legorburu I, Bald J, Chust G, Ruiz-Minguela P, et al. A marine spatial planning approach to select suitable areas for installing wave energy

converters (WECs), on the Basque Continental Shelf (Bay of Biscay). Coast Manag 2012;40:1–19. http://dx.doi.org/10.1080/08920753.2011.637483.

- [129] Azzellino A, Kofoed JP, Lanfredi C, Margheritini L, Pedersen ML. A Marine Spatial Planning framework for the optimal siting of Marine Renewable Energy Installations: two Danish case studies. J Coast Res 2013;29:1623–8. http://dx.doi.org/10.2112/SI65-274.1.
- [130] Corsatea TD. Increasing synergies between institutions and technology developers: Lessons from marine energy. Energy Policy 2014:1–16. <u>http://dx.</u> doi.org/10.1016/j.enpol.2014.07.006.
- [131] Wright G. Regulating marine renewable energy: a brief literature review; 2012. (http://www.glenwright.net/regulating-marine-renewable-energy-abrief-literature-review/#.U0UPvhBLpzp).
- [132] Corsatea TD, Magagna D. Overview of European innovation activities in marine energy technology. Luxembourg: Publications Office of the European Union; <u>http://dx.doi.org/10.2790/99213</u>.
- [133] Van Nieuwkoop JCC, Smith HCM, Smith GH, Johanning L. Wave resource assessment along the Cornish coast (UK) from a 23-year hindcast dataset validated against buoy measurements. Renew Energy 2013;58:1–14. <u>http:</u> //dx.doi.org/10.1016/j.renene.2013.02.033.
- [134] Ayat B. Wave power atlas of Eastern Mediterranean and Aegean Seas. Energy 2013;54:251–62. <u>http://dx.doi.org/10.1016/j.energy.2013.02.060</u>.
- [135] Rusu L, Guedes Soares C. Wave energy assessments in the Azores islands. Renew Energy 2012;45:183–96. <u>http://dx.doi.org/10.1016/j.renene.2012.02.027</u>.
- [136] Stopa JE, Filipot J-F, Li N, Cheung KF, Chen Y-L, Vega L. Wave energy resources along the Hawaiian Island chain. Renew Energy 2013;55:305–21. <u>http://dx. doi.org/10.1016/j.renene.2012.12.030.</u>
- [137] Sandhya KG, Balakrishnan Nair TM, Bhaskaran PK, Sabique L, Arun N, Jeykumar K. Wave forecasting system for operational use and its validation at coastal Puducherry, east coast of India. Ocean Eng 2014;80:64–72. <u>http://dx. doi.org/10.1016/j.oceaneng.2014.01.009</u>.
- [138] Goundar JN, Ahmed MR. Marine current energy resource assessment and design of a marine current turbine for Fiji. Renew Energy 2014;65:14–22. http://dx.doi.org/10.1016/j.renene.2013.06.036.
- [139] Rashid A. Status and potentials of tidal in-stream energy resources in the southern coasts of Iran: a case study. Renew Sustain Energy Rev 2012;16:6668–77. http://dx.doi.org/10.1016/j.rser.2012.08.010.
- [140] Blunden LS, Bahaj AS, Aziz NS. Tidal current power for Indonesia? An initial resource estimation for the Alas Strait Renew Energy 2013;49:137–42. <u>http:</u> //dx.doi.org/10.1016/j.renene.2012.01.046.
- [141] Sutherland G, Foreman M, Garrett C. Tidal current energy assessment for Johnstone Strait, Vancouver Island. Proc Inst Mech Eng Part A J Power Energy 2007;221:147–57. http://dx.doi.org/10.1243/09576509JPE338.
- [142] Grabbe M, Lalander E, Lundin S, Leijon M. A review of the tidal current energy resource in Norway. Renew Sustain Energy Rev 2009;13:1898–909. http://dx.doi.org/10.1016/j.rser.2009.01.026.
- [143] Karsten RH, McMillan JM, Lickley MJ, Haynes RD. Assessment of tidal current energy in the Minas Passage, Bay of Fundy. Proc Inst Mech Eng Part A J Power Energy 2008;222:493–507. http://dx.doi.org/10.1243/09576509JPE555.
- [144] Chen W-B, Liu W-C, Hsu M-H. Modeling assessment of tidal current energy at Kinmen Island, Taiwan. Renew Energy 2013;50:1073–82. <u>http://dx.doi.org/</u> 10.1016/j.renene.2012.08.080.
- [145] Work PA, Haas KA, Defne Z, Gay T. Tidal stream energy site assessment via three-dimensional model and measurements. Appl Energy 2013;102:510–9. http://dx.doi.org/10.1016/j.apenergy.2012.08.040.
- [146] Zhang J-S, Wang J, Tao A-F, Zheng J-H, Li H. New concept for assessment of tidal current energy in Jiangsu Coast, China. Adv Mech Eng 2013;2013(9). http://dx.doi.org/10.1155/2013/340501.
- [147] Lawless M, Rodger D. Development of the European tidal database and its potential application to marine renewables. J Coast Res 2013;29:1629–30. http://dx.doi.org/10.2112/SI65-275.1.
- [148] Sørensen HC, Naef S, Anderberg S, Hauschild MZ. Life cycle assessment of the wave energy converter: wave dragon. In: Schmid J, editor. International conference on ocean energy - from Innov. to Ind., Bremerhaven; 2006.
- [149] Sørensen HC, Naef S. Report on technical specification of reference technologies (wave and tidal power plant); 2008.
- [150] Parker RPM, Harrison GP, Chick JP. Energy and carbon audit of an offshore wave energy converter. Proc Inst Mech Eng Part A J Power Energy 2007;221:1119–30. <u>http://dx.doi.org/10.1243/09576509JPE483</u>.
- [151] Thomson RC, Harrison GP, Chick JP. Full life cycle assessment of a wave energy converter. In: IET conference on renewable power generation (RPG 2011), Edinburgh: IET; 2011, p. 63. doi:10.1049/cp.2011.0124.
- [152] Walker S, Howell R. Life cycle comparison of a wave and tidal energy device. Proc Inst Mech Eng Part M J Eng Marit Environ 2011;225:325–37. <u>http://dx.doi.org/10.1177/1475090211418892</u>.
- [153] Douglas CA, Harrison GP, Chick JP. Life cycle assessment of the Seagen marine current turbine. Proc Inst Mech Eng Part M J Eng Marit Environ 2008;222:1– 12. http://dx.doi.org/10.1243/14750902JEME94.